Author's response to reviews

Title: Ultrasound imaging versus morphopathology in cardiovascular diseases. Coronary collateral circulation and atherosclerotic plaque

Authors:

Giorgio Baroldi (Riccardo.Bigi@unimi.it)
Riccardo Bigi (Riccardo.Bigi@unimi.it)
Lauro Cortigiani (lacortig@tin.it)

Version: 2 Date: 28 February 2005

Author's response to reviews:

Eugenio Picano MD, PhD
Editor, Cardiovascular Ultrasound

Please find enclosed the revised version of the manuscript by Baroldi G. et Al. entitled "Ultrasound imaging versus morphopathology in cardiovascular diseases. Coronary collateral circulation and atherosclerotic plaque". The Authors would like to thank you and the Reviewers for the useful suggestions and for giving us the opportunity to improve the overall quality of this paper. All comments and suggestions have been carefully addressed and the manuscript has been modified accordingly. We do hope that the manuscript may be suitable for publication in this revised form.

Best Regards,

Prof. Giorgio Baroldi